Author's response to reviews

Title: Gatekeepers of health: a qualitative assessment of child care centre staff's perspectives, practices and challenges to enteric illness prevention and management in child care centres

Authors:

Marsha Taylor (marsha_taylor@phac-aspc.gc.ca)
Cindy L Adams (cadams@ucalgary.ca)
Andrea Ellis (andrea_ellis@phac-aspc.gc.ca)

Version: 8 Date: 11 June 2008

Author's response to reviews:

A minor change was made to incorporate the text of a footnote into the body of the manuscript. I was unable to make this change through the proofing stage - email sent June 10, 2008.

No further changes to the content were made. I consider this edit to be part of final proofing and confirm publication of the revised manuscript.